Author's response to reviews

Title: Virtual microscopy system at Chinese medical university: an assisted teaching platform for promoting active learning and problem-solving skills

Authors:

Yanping Tian Dr. (tianyp1981@163.com)
Wengang Xiao Dr. (342514806@qq.com)
Chengren Li Prof. (tianyp1981@aliyun.com)
Yunlai Liu Prof. (tianyp1981@tmmu.edu.cn)
Maolin Qin Dr. (wanghuan_1357@163.com)
Yi Wu Dr. (wanghuan_135@163.com)
Lan Xiao Prof. (363309136@qq.com)
Hongli Li Prof. (lihlimm@163.com)

Version: 3 Date: 27 March 2014

Author's response to reviews: see over
MS: 7424293941119609
Research article
Virtual Microscopy System at Chinese Medical University: An Assisted Teaching Platform for promoting active learning and problem-solving skills
Yanping Tian Dr., Wengang Xiao Dr., Chengren Li Prof., Yunlai Liu Prof., Maolin Qin Dr., Yi Wu Dr., Lan Xiao Prof. and Hongli Li Prof.
BMC Medical Education

Dear editors

On behalf of all my colleagues, I’d like to thank you very much for your positive decision on our manuscript. We found that both your and the referees’ comments are constructive and helpful. Edanz group China was hired for a professional editing service (Edited manuscript number: C1403-32552-TianYanping). The amendments were showed colored text in the manuscript.

(http://www.liwenbianji.cn/bmc_welcome?utm_source=bmc&utm_medium=marketing&utm_content=global_link&campaign=bmc_general)

理文编缉
Edanz Group China

We hope we have addressed all the questions and are willing to re-revise if we were asked to do so.

Looking forward to hearing from you.

Yours sincerely,
Hongli Li MD
Department of Histology and Embryology,
School of basic medicine,
Third Military Medical University
Chongqing 400038,
PR China
Tel: 86-23-68752219